\documentclass[]{adamjoucc}

%\usepackage{color}
%...


% ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM 
%%For Journal Use Only, Do Not Modify
%\journalnumber{nn}
%\journalyear{year}
%\journalpages{1}{x}
%\oddrunninghead{A. Uthor: Title}
%\dates{day month year}{xx xx xx}{xx xx xx}
%% ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM ADAM


\begin{document}



\begin{frontmatter}   %%  Title, information about author, abstract, etc.

\titledata{Title of the paper}           % title of the paper
{Footnote on the title.}                 % footnote on the title -- empty if not required

\authordata{First A. Author}            % First author name
{Institution, Address, City, Country}    % Affiliation and address
{email@address1.com}                     % E-mail address
{Footnote on the first author.}          % Footnote on the first author (grant number, thanks, 
                                         % web page, etc.) -- empty in not required

\authordata{Second B. Author}            % Second author
{Institution, Address, City, Country} 
{email@address2.com}
{}                                       % No footnote!

\authordata{Third. C. Author}            % Third author
{Institution, Address, City, Country}
{email@address3.com}
{Footnote on the third author.}

%%%%%%% Two authors from the same institution 

%\authordatatwo{First A. Author}{email@address1.com}{First Author footnote, leave blank if none}%footnote for first author
%{Second B. Author}{email@address2.com}{second Author footnote, leave blank if none}
%{Institution, Address, City, Country}

%%%%%%%%% Three authors same institution

%\begin{center}
%\authordatanoaffil{First A. Author}
%{email@address1.com}
%{},
%\authordatanoaffil{Second B. Author}
%{email@address1.com}
%{},
%\authordatanoaffil{Third. C. Author}
%{email@address1.com}
%{corresponding author}\\[1mm]
%\affiliation{Institution, Address, City, Country}
%\end{center}


\keywords{Groups, graphs.}               % Keywords
\msc{05C15, 05C10}                       % Math. Subj. Class. codes

\begin{abstract}
The abstract should state the problem studied and the principal results proved. It should be concise, it should not include complicated formulas and embedded equations, and mathematical formulae should be kept to a minimum. (The abstract needs to be able to be converted to HTML for posting on the website!) It should not include explicit citations.

The abstract should not contain citations or references to \LaTeX labels used in the document, nor should it contain complicated mathematical formatting.
\end{abstract}

\end{frontmatter}   %% End of the front matter

%% Your article

\section{The first section}

Text in the first section. Text in the first section. Text in the first section.
Text in the first section. Text in the first section. Text in the first section.
Text in the first section. Text in the first section. Text in the first section.

All references to theorems, figures, sections, etc. should be handled by \verb=\label{....}= and \verb=\ref{...}= calls. For example, in Section \ref{secondSec}, we include some text, while in Sections \ref{theorems} and \ref{moreThms} we give examples of our theorem environments. In the proof of Theorem \ref{mainThm}, we use Definition \ref{mydef} to complete the proof. Finally, in Figure \ref{rect} we show a very small rectangle.

\section{The second section}\label{secondSec}

Text in the second section. Text in the second section. Text in the second section.
Text in the second section. Text in the second section. Text in the second section.
Text in the second section. Text in the second section. Text in the second section.

\subsection{A small subsection} Text in the subsection.
\section{Theorem environments}\label{theorems}

The class has the following major environments defined.

\begin{defn}  \label{mydef}                         %% Environment for definition
Definition.
\end{defn}

\begin{thm}          \label{mainThm}                   %% Environment for theorem
The first theorem.
\begin{proof}
Following the suggestion presented in \cite{BI1}, we apply Definition \ref{mydef} to observe some stuff. This allows us to complete the proof of the first theorem. Notice we're using a built-in proof environment. Please do the same!
\end{proof}
\end{thm}
                                        %% More environments...
\section{More theorem-like environments}\label{moreThms}
The following theorem-like environments are pre-defined.
%\begin{enumerate}
%\item \texttt{thm} (Theorem)
%\item \texttt{prop} (Proposition)
%\item \texttt{lem} (Lemma)
%\item \texttt{cor} (Corollary)
%\item \texttt{claim} (Claim)
%\item \texttt{axiom} (Axiom)
%\item \texttt{conj} (Conjecture)
%\item \texttt{hypo} (Hypothesis)
%\item \texttt{assum} (Assumption)
%\end{enumerate}
\begin{verbatim}
\theoremstyle{plain}
\newtheorem{thm}{Theorem}[section]
\newtheorem{theorem}[thm]{Theorem}
\newtheorem{prop}[thm]{Proposition}
\newtheorem{proposition}[thm]{Proposition}
\newtheorem{lem}[thm]{Lemma}
\newtheorem{lemma}[thm]{Lemma}
\newtheorem{cor}[thm]{Corollary}
\newtheorem{corollary}[thm]{Corollary}
\newtheorem{claim}[thm]{Claim}
\newtheorem{axiom}[thm]{Axiom}
\newtheorem{conj}[thm]{Conjecture}
\newtheorem{conjecture}[thm]{Conjecture}
\newtheorem{fact}[thm]{Fact}
\newtheorem{hypo}[thm]{Hypothesis}
\newtheorem{assum}[thm]{Assumption}
\newtheorem{crit}[thm]{Criterion}
\newtheorem{cons}[thm]{Construction}
\end{verbatim}

The following definition-like environments can be used.
%\begin{enumerate}
%\item \texttt{defn} (Definition)
%\item \texttt{exmp} (Example)
%\item \texttt{rem} (Remark)
%\item \texttt{prob} (Problem)
%\item \texttt{quest} (Question)
%\item \texttt{alg} (Algorithm)
%\end{enumerate}

\begin{verbatim}
\theoremstyle{definition}
\newtheorem{defn}[thm]{Definition}
\newtheorem{exmp}[thm]{Example}
\newtheorem{example}[thm]{Example}
\newtheorem{rem}[thm]{Remark}
\newtheorem{remark}[thm]{Remark}
\newtheorem{prob}[thm]{Problem}
\newtheorem{problem}[thm]{Problem}
\newtheorem{prin}[thm]{Principle}
\newtheorem{quest}[thm]{Question}
\newtheorem{alg}[thm]{Algorithm}

\end{verbatim}

\section{Figures}

In this section, Figure \ref{rect} shows a small rectangle. When embedding graphics, please provide them either as .pdf files or as TiKz figures. If at all possible, do not use .eps or .jpg graphics.

\begin{figure}[htbp]
\begin{center}
$\Box$
\caption{A very small rectangle.}
\label{rect}
\end{center}
\end{figure}


%% Example figure
%\begin{figure}[htbp]
%\begin{center}
%\includegraphics[width=.4\linewidth]{mypic}
%\caption{A very small rectangle.}
%\label{rect}
%\end{center}
%\end{figure}


%% Bibliography

\section{References}
Examples of references:~\cite{BI1}, \cite{BI2}, \cite{BI3} and \cite{BI6}.
\nocite{BI4,BI5} Note that you should cite every reference listed in your list of references!

%Note that we still need to fix the formatting of electronic references in the journal style file .

Note that the small red numbers that appear after the references are intentional and indicate the page number on which the reference citation appears!

\bibliographystyle{adamjoucc}
\bibliography{adamexample}


%\begin{thebibliography}{99}
%
%\bibitem{BI1}
%First A. Author, Second B. Author and Third C. Author, 
%Article title, 
%\emph{Journal Title} \textbf{1} (1982), 1--100.
%
%\bibitem{BI2}
%First A. Author, 
%\emph{Book title}, third ed., 
%Publisher, New York, 1982.
%
%\bibitem{BI3}
%First A. Author and Second B. Author, 
%Chapter in an edited book, 
%in: First Editor, Second Editor (eds.), 
%\emph{Book Title},
%Publisher, Amsterdam, 1999, 232--345.
%
%\end{thebibliography}

\end{document}
